Author’s response to reviews

Title: Breast metastasis from a squamous cells carcinoma of the oropharynx: description of a case report

Authors:

Raffaele Longo (raflongo@libero.it)
Emmanuelle Melgar (Emmanuelle-melgar@hotmail.fr)
Marco Campitiello (m.campitiello@chr-metz-thionville.fr)
Francesca Plastino (f.plastino@chr-metz-thionville.fr)
Nada Eid (n.eid@chr-metz-thionville.fr)
Isabelle Quirin (i.quirin@chr-metz-thionville.fr)
Laurent Hennequin (l.hennequin@chr-metz-thionville.fr)
Yves Grignon (y.grignon@chr-metz-thionville.fr)
Michel Gunther (m.gunther@chr-metz-thionville.fr)
Philippe Quetin (p.quetin@chr-metz-thionville.fr)

Version: 1 Date: 13 Oct 2017

Author’s response to reviews:

JMCR-D-17-00400

Breast metastasis from a squamous cells carcinoma of the oropharynx: description of a case report

Raffaele Longo, M.D.; Emmanuelle Melgar; Marco Campitiello; Francesca Plastino; Nada Eid; Isabelle Quirin; Laurent Hennequin; Yves Grignon; Michel Gunther; Philippe Quetin

Journal of Medical Case Reports
Dear Editor,

Please find the final changes of the m/s as suggested.

I would like to thank the Reviewers for their comments.

Reviewer reports:

Reviewer #1:

2. Given the patient has died, the consenement must be obtained from the patient's close relatives.

* Page 7 lines 12 - 14: "Written informed consent was obtained from the patient for publication of this Case report and any accompanying images".

A- Patient’s close relatives gave written, informed consent to publish this case and any accompanying images. This aspect was highlighted in the appropriate section of the m/s.

Reviewer #2:

9. Additional comments for the author(s)? The last images (Figures 1D and 1E) with the histological pictures are very poor, at low power and with low resolution. I would suggest new pictures, including higher power (20x or 40x objectives) to reveal some of the cytological features of the tumor, as described in the text for these figures.

A- The Figures have been replaced with higher power and resolution images as suggested.

Yours sincerely,

Dr R. Longo